Author's response to reviews

Title: Reliability and validity of CDAI and SDAI indices in comparison to DAS-28 index in Moroccan patients with rheumatoid arthritis

Authors:

Imane Ben Slama (benslamaimane@gmail.com)
Fadoua Allali (fadouaallali@yahoo.fr)
Touria Lakhdar (ltouria@hotmail.fr)
Sarra El kabbaj (k_sarra99@hotmail.fr)
Lamyae Medrare (lamyamedrar@gmail.com)
Ange Ngeuleu (ngeuleuan@gmail.fr)
Hanan Rkain (hanankain@yahoo.fr)
Najia Hajjaj-Hassouni (nhajjajhassouni@gmail.com)

Version: 5
Date: 21 August 2015

Author's response to reviews:

- the corresponding author is indicated in the title page
- The abstract is on a separate page to the title page
- The abstract is according to the Instructions for Authors:
- section Patients, materials and methods is renamed as Methods
- Figure legends are placed at the end of the main manuscript and not within the figure files.